

Indicators for monitoring performance (RP4)

Final report



June 2024

Performance Review Body of the Single European Sky | Rue de la Fusée 96, Office 50.659, 1130 Brussels Office Telephone: +32 (0)2 234 7824 | cathy.mannion@prb.eusinglesky.eu | prb-office@prb.eusinglesky.eu | eu-single-sky.transport.ec.europa.eu

COPYRIGHT NOTICE	© European Union, 2024
AND DISCLAIMER	This report has been prepared for the European Commission by the Performance Review Body of the
	Single European Sky (PRB).
	Reproduction is authorised provided the source is acknowledged. However, neither the European Com-
	mission, nor any person acting on its behalf, may be held responsible for the use which may be made of
	the information contained in this publication, or for any errors which may appear, despite careful prep-
	aration and checking.

TABLE OF CONTENTS

1	INTRODUCTION	. 2
2	SAFETY	. 3
2.1	Existing indicators within the Regulation	. 3
2.2	Review of Steer proposals for existing indicators	. 3
2.3	Review of Steer and EASA proposals for new indicators	. 6
2.4	Option 1 – Prescriptive PIs based on occurrences	. 6
2.5	Option 2 – EASA revised proposed new indicators	. 7
2.6	PRB suggestions for new indicators	. 7
3	ENVIRONMENT	. 9
3.1	Existing indicators within the Regulation	. 9
3.2	Review of Steer proposals for existing indicators	. 9
3.3	Review of Steer proposals for new indicators	13
3.4	PRB suggestions for new indicators	14
4	CAPACITY	17
4.1	Existing indicators within the Regulation	17
4.2	Review of Steer proposals for existing indicators	17
4.3	Review of Steer proposals for new indicators	20
4.4	PRB suggestions for additional new indicators	21
5	COST-EFFICIENCY	24
5.1	Existing indicators within the Regulation	24
5.2	Review of Steer proposals for existing indicators	24
5.3	Review of Steer proposals for new indicators	26
5.4	PRB suggestions for additional new indicators	26
6	NETWORK MANAGER	27
6.1	Safety indicators	27
6.2	Environment indicators	27
6.3	Capacity indicators	27
6.4	Cost-efficiency indicators	29
7	CONCLUSIONS	30

1 INTRODUCTION

- 1 The Commission contracted Steer to complete an independent review of the indicators as defined in in the Commission Implementing Regulation (EU) 2019/317 (hereafter the Regulation) in view of the fourth reference period (RP4). Steer conducted bilateral interviews, collected a questionnaire, and organised a stakeholder workshop to gather views from stakeholders on the topic. This resulted in a report to the Commission which proposes a way forward for indicators within each of the key performance areas (KPAs) of the Regulation.
- 2 One of the primary tasks of the Performance Review Body (PRB) is to assist the Commission when defining the key performance indicators and indicators for monitoring under Article 3(c) of the Regulation.
- An initial report was distributed to the participant 3 of the for the technical meeting with the stakeholders which took place on 23rd May. This report is the PRB's final report and includes the updates following the technical meeting. The scope of this report is to suggest monitoring indicators that should be implemented for the fourth reference period of the performance and charging scheme, some of which may be considered as key performance indicators in future reference periods. In its assessment, the PRB has assumed that key performance indicators will remain the same for RP4 as they are in RP3 and are therefore out of scope. As part of this exercise, the PRB has considered the proposals included in Steer's report which represent an analysis of stakeholders' views.
- ⁴ The objective of the PRB is to take stock of the experience gained during the current and earlier reference periods in order to refine the existing set of monitoring indicators, and where necessary add complementary indicators to provide greater depth in the assessment of ATM performance for the start of RP4. The main objectives of the indicators suggested are:
 - Safety: Include non-prescriptive and flexible monitoring indicators to allow for analysis of local high-risk areas. The proposal broadens the safety monitoring to include

interdependencies with other key performance areas, in line with the work in progress for the preparation of RP5.

- Environment: Increase the importance of environment monitoring by covering all the flight phases and including indicators on fuel consumption. The suggested indicators have the advantage of potentially forming the basis for new key performance indicators for RP5; allowing for a more focused assessment of ATM performance.
- Capacity: Complement the key performance indicator on average delays by including monitoring indicators on capacity provision and measures of interdependencies with other key performance areas.
- Cost-efficiency: Include indicators on interdependencies with other key performance areas. and
- Network manager function: Rationalise existing indicators.
- 5 The PRB considers that the suggested new performance indicators, included in this paper, are important first steps in an evolutionary process. The PRB proposes to carry on this process with further studies focused on developing more refined indicators and key performance indicators that better reflect the impact of actions taken by ANSPs, for inclusion later in RP4 and for consideration as KPIs in RP5.
- 6 This report is structured as follows:
 - Section 2 presents the suggestions related to the indicators for the safety key performance area (KPA);
 - Section 3 presents the suggestions related to the indicators for the environment KPA;
 - Section 4 presents the suggestions related to the indicators for the capacity KPA;
 - Section 5 presents the suggestions related to the indicators for the cost-efficiency KPA;
 - Section 6 presents the suggestions related to the network functions indicators; and
 - Section 7 presents a summary.

2.1 Existing indicators within the Regulation

- 7 Under Annex I of the Regulation, the following KPI and PIs are used to assess the performance within the safety KPA at Union-wide level:
 - Minimum level of the Effectiveness of Safety Management (EoSM) (KPI) (Section 1, 1.1);
 - The rate of runway incursions at Union-wide level with a safety impact (Section 1, 1.2(a)); and
 - The rate of separation minima infringements at Union-wide level with a safety impact (Section 1, 1.2(b)).
- 8 For the local level, the following KPI and PIs are used to assess the performance within the safety KPA:
 - Minimum level of the Effectiveness of Safety Management (KPI) (Section 2, 1.1);
 - The rate of runway incursions at local level (Member State) (Section 2, 1.2(a));
 - The rate of separation minima infringements within the airspace of all controlling air traffic services units at local level (Member State) (Section 2, 1.2(b));
 - The rate of runway incursions with ATS/CNS contribution at local level (airport) (Section 2, 1.2(c));
 - The rate of separation minima infringements with ATS/CNS contribution at local level (ANSP) (Section 2, 2.1(d)); and
 - Application by the ANSPs of automated safety data recording systems (Section 2, 2.1(e)).

2.2 Review of Steer proposals for existing indicators

9 Steer proposes retaining all existing safety performance indicators. Table 1 (next pages) summarises the Steer proposals and the PRB suggestions regarding the existing safety indicators.

Minimum level of the effectiveness of Safety Management at Union-wide and local levels (EoSM)

10 The EoSM is the sole KPI for the safety KPA. It is a process indicator which measures the minimum

level of maturity achieved within five different safety management objectives:

- Safety policy and objectives;
- Safety risk management;
- Safety assurance;
- Safety promotion; and
- Safety culture.
- 11 The minimum level of maturity is determined based on the EoSM questionnaire.
- ¹² Steer suggests retaining the EoSM as the sole KPI in line with the recommendation from the EASA S(K)PI drafting group and proposes to revise the EoSM questionnaire in preparation for RP4. Steer notes that the KPI has been used throughout the reference periods and is a well-known indicator for ANSPs, Network Manager, and NSAs. Steer notes that the KPI does not necessarily measure actual safety performance, it measures the level of implementation of the Safety Management System.
- 13 The PRB highlights that the revised RP4 EoSM is already available and has been consulted with stakeholders in a process managed by EASA.

Rate of runway incursions with a safety impact at Union-wide level

- 14 This indicator is defined as the total number of runway incursions with a safety impact that occurred at regulated airports in all Member States divided by the total number of IFR and VFR airport movements.¹
- Steer suggests retaining this indicator at the Union-wide level for monitoring as an outcome indicator. Steer also suggests a potential refinement by grouping airports with a similar type or level of traffic for the reporting. However, Steer does not develop the proposal in detail and does not specify if the grouping would apply to all indicators measuring rates of runway incursions. Finally, Steer suggests enhancing the data reported to better analyse the implications but does not further develop the proposal.

¹ Occurrences with safety impact should be understood as those occurrences that may represent a risk to aviation. The way to identify these types of occurrences is using the safety risk grade red or amber in the European Risk Classification Scheme (ERCS) matrix when applied to SMIs and RIs, and the ground severity classification A, B, or C after applying the risk analysis tool (RAT) to SMIs and RIs with ATS/CNS contribution.

The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. The PRB supports Steer's proposal to group airports, considering that drawing conclusions from comparisons between different reporting organisations can be challenging. This analysis can be done based on the data currently available.

Rate of separation minima infringements at Unionwide level with a safety impact

- 17 This indicator is defined as the Union-wide total number of separation minima infringements with a safety impact divided by the Union-wide controlled IFR flight hours within the airspace.
- Steer suggests retaining this indicator at the Union-wide level for monitoring as an outcome indicator. In addition, Steer proposes to enhance the data reported to better analyse the implications but does not further develop the proposal.
- 19 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

Rate of runway incursions at local level (Member State)

- 20 This indicator is defined as the total number of runway incursions with a safety impact that occurred at regulated airports in a Member State divided by the total number of IFR and VFR airport movements.
- Steer suggests retaining the indicator at the Member State level for monitoring as an outcome indicator. Similarly for the Union-wide level indicator, Steer also suggests a potential refinement by grouping airports with similar types/levels of traffic for the reporting, and to enhance the data reported to better analyse differences between group of airports. Steer has not developed further the definition of proposed airport groups.
- ²² The PRB agrees with Steer's proposal and suggests retaining this indicator. PRB supports analysing groups of similar airports (this can be achieved during the monitoring process without further data requests).

Rate of separation minima infringements within the airspace of all controlling air traffic services units at local level (Member State)

23 This indicator is defined as the total number of separation minima infringements with a safety impact that occurred within the airspace of all air traffic service units in a Member State divided by the total number of controlled IFR flight hours within the respective airspace.

- Steer suggests retaining the indicator at the Member State level as a lagging indicator measuring actual safety performance. As for the Union-wide indicator, Steer proposes to enhance the data reported to better analyse the implications but does not further develop the proposal.
- ²⁵ The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

Rate of runway incursions with ATS/CNS contribution at local level (airport)

- ²⁶ This indicator is defined as the total number of runway incursions with a safety impact that have any contribution from air traffic or CNS services at a specific airport divided by the total number of IFR and VFR movements at that airport.
- 27 Steer suggests retaining this indicator at the Member State level for monitoring as an outcome indicator. As for the Union-wide indicator, Steer proposes to enhance the data reported to better analyse the implications but does not further develop the proposal.
- 28 The PRB agrees with Steer's proposal and suggests retaining the indicator for RP4. The PRB supports analysing groups of similar airports (this can be achieved during the monitoring process without further data requests).

Rate of separation minima infringements with ATS/CNS contribution at local level (ANSP)

- 29 This indicator is defined as the total number of separation minima infringements with a safety impact that have any contribution from air traffic or CNS (communications, navigation and surveillance) services divided by the total number of controlled IFR flight hours within the airspace controlled by the air navigation service provider.
- Steer suggests retaining this indicator at the local level, monitoring it as an outcome indicator. As for the Union-wide indicator, Steer proposes to enhance the data reported to better analyse the implications but does not further develop the proposal.
- The PRB agrees with Steer's proposal and suggests retaining the indicator for RP4.

Application by the ANSPs of automated safety data recording systems

- ³² This indicator defines whether ANSPs use automated safety data recording tools to improve the gathering of occurrence data (SMI and RIs) and the analysis by the organisations' safety management system (SMS).
- 33 Steer suggests retaining this indicator at the local level, without proposing any further changes.
- The PRB highlights that the indicator has shown little improvement in the use of such systems over RP2 and RP3 and does not support the performance monitoring as originally intended. Therefore, the PRB (and EASA) suggests removing this indicator.

	Scope				
Indicator	Union- wide	Local	Steer proposal	PRB suggestion	
Minimum level of the Effective- ness of Safety Management (EoSM) (KPI)	~	~	No change, revise EoSM.	No change, revise EoSM.	
The rate of runway incursions at Union-wide level with a safety impact	✓		No change.	No change. Grouping can be done based on already re- ported data.	
The rate of separation minima infringements at Union-wide level with a safety impact	~		No change.	No change.	
The rate of runway incursions at local level (Member State)		\checkmark	No change.	No change. Grouping can be done based on already re- ported data.	
The rate of separation minima infringements within the air- space of all controlling air traffic services units at local level (Member State)		~	No change.	No change.	
The rate of runway incursions with ATS/CNS contribution at lo- cal level (airport)		~	No change.	No change. Grouping can be done based on already re- ported data.	
The rate of separation minima infringements with ATS/CNS contribution at local level (ANSP)		~	No change.	No change.	
Application by the ANSPs of au- tomated safety data recording systems		~	No change.	Remove. The indicator has little value for safety perfor- mance monitoring.	

Table 1 - Summary table of the current safety indicators.

2.3 Review of Steer and EASA proposals for new indicators

- ³⁵ In addition to the existing indicators, Steer proposes two options for new safety performance indicators to be introduced in RP4.
- ³⁶ The first option is to introduce a selection of prescriptive safety performance indicators:
 - Number of overloads reported by controllers (local level);
 - ATM-specific occurrences (Union-wide and local levels);
 - Airspace infringements (Union-wide and local levels); and
 - Loss of minimum safety altitudes (Union-wide and local levels).
- ³⁷ The second option was based on a proposal by EASA to include a new set of indicators related to the risk picture at Union-wide, Member State, and ANSP levels. Following stakeholders' comments at the technical meeting on 23rd May, EASA revised the proposal by only including indicators related to the risk picture at ANSP levels:
 - Locally identified indicators in the ANSP Management System (local level).
- Table 2 (next pages) summarises the first option proposed by Steer, EASA's revised proposal (replacing Option 2 as set out in the Steer report), and the PRB suggestions regarding the new safety indicators for RP4.

2.4 Option 1 – Prescriptive PIs based on occurrences

Number of overloads reported by controllers

- 39 Steer suggests introducing the overload occurrences reported by the controllers as a local level indicator. However, Steer does not specify if the indicator should be defined as a number or as a rate, and whether occurrences with a safety impact should only be reported.
- 40 The indicator identifies situations where the controller judges that they are required to handle more traffic than considered safe. This indicator monitors the ability to safely manage increased complexity, increased re-routings, signs of stress and fatigue, definition of sector capacity and ability to protect controllers through flow measures.

In terms of data sources, overload occurrences are reportable events under Implementing Regulation (EU) 2015/1018 Annex III, 2(8) or 3(7). Hence, additional effort in reporting on this indicator should be minimal in terms of collection of existing occurrence reports. The PRB suggests enhancing the data reported for analytic reasons by requesting Member States to provide a list of the occurrences which are the basis for the rate/number calculation (e.g. if one ATS unit is more prone to overload than other ATS units).

ATM-specific occurrences

- 42 Steer suggests introducing the monitoring of ATM-specific occurrences at Union-wide and local levels. This safety indicator was monitored during RP2. However, Steer does not specify if the indicator should be defined as a number or as a rate, and whether occurrences with a safety impact should only be reported.
- 43 This indicator monitors the level of degradation or total loss of services or functions (e.g. inability to provide ATM services or to execute ATM functions due to missing, significantly incorrect, corrupted, inadequate, or misleading information from any support service, etc). It would therefore also include aspects such as technical resilience, cyberattacks, and GNSS jamming affecting navigational aids however, this level of granularity is not provided by the suggested indicator.
- In terms of data sources, the ATM-specific occurrences are reportable events under Implementing Regulation (EU) 2015/1018 Annex III, 2, and 3 (e.g. points (2) and (3)). Hence, additional effort in reporting on this indicator should be minimal in terms of the collection of existing occurrence reports.

Airspace infringements

- 45 Steer suggests introducing the monitoring of airspace infringements at Union-wide and local levels. This safety indicator was monitored during RP2. However, Steer does not specify if the indicator should be defined as a number or as a rate, and whether occurrences with a safety impact should only be reported.
- ⁴⁶ This indicator identifies aspects such as infringements caused by new, high and low-level airspace entrants and unidentified aircraft (those with their transponder switched off).

- 47 This indicator monitors airspace infringements, in particular whether the occurrences increase over time for example due to Remotely Piloted Aircraft Systems, increased use of and/or more dynamic reserved airspace principles.
- In terms of data sources, airspace infringements are reportable events under Implementing Regulation (EU) 2015/1018, Annex III, 1(10b). Hence, additional effort in reporting on this indicator should be minimal to collect existing occurrence reports.

Rate of loss of Minimum Safe Altitudes

49 Steer suggests introducing an indicator covering loss of minimum safe altitudes at Union-wide and local levels. The indicator would cover safety issues related to performance-based navigation (PBN)/required navigation performance (RNP) operations which are increasing in use. Steer reports that the data is already collected, and, while the ANSPs are not fully in control of such occurrences, they are a contributory/related factor.

2.5 Option 2 – EASA revised proposed new indicators

- In its revised option, EASA argues that adding new prescriptive indicators on occurrences provides limited value to safety performance monitoring and may damage occurrence reporting by incentivising low reporting rates. EASA also notes that prescriptive indicators may divert the attention and resources of ANSPs and NSAs to monitoring and analysing these indicators and not to local high-risk safety issues. Therefore, EASA proposes a shift of paradigm: From a centrally prescribed indicators towards monitoring the indicators that ANSPs have identified as being of highest risk.
- 51 EASA's revised proposal makes use of the existing authority and organisation requirements to monitor safety. It covers safety performance indicators that are numerically calculated and grouped to produce a qualitative view of the risk picture, which can be compared. Data used to derive the risk picture may change from year to year as risks are mitigated and new risks emerge.

Locally identified indicators in the ANSP Management System (local level)

⁵² Through this indicator, ANSPs would report on the indicators they are monitoring, and which reflect locally defined high risks. As per Regulation (EU)

2017/373, Subpart (ATM/ Annex III, В ANS.OR.B.005), ANSPs are required to describe in their management systems the means to verify the performance of the service provider's organisation considering the performance indicators and performance targets of the management system. Performance indicators are described in the AMC (acceptable means of compliance) and examples further developed in the GM (guidance material). The indicators defined can also include business indicators (e.g. ATCO staffing levels, funding issues) if they influence the risk picture.

- ⁵³ The monitoring covers indicators of relevance to ANS/ATM performance and the overall business health of the service provider. The full range of indicators that are monitored by the business should be considered, with those exhibiting the highest risk to the provision of service being chosen to form the risk picture. Examples of information that could be provided are:
 - The reasons why the indicator was selected for monitoring at ANSP level;
 - A visualisation of the development of the indicator over time; and
 - An assessment of the significance of the data.

The indicator could be supported by explanations of why the set of locally defined indicators can be used to develop the risk picture, and what developments have been seen compared to the situation at the time the performance plan(s) were adopted. The evolution of the risk picture might then be used to monitor the emergence of new high-risk areas and actions taken to monitor/address new high risks.

- 54 The value of the indicator is dependent on the maturity of the ANSPs management system (i.e. implementation of regulatory requirement) and represents an opportunity to compare annual risk pictures with those contained in Performance Plans.
- 2.6 PRB suggestions for new indicators
- 55 Prescriptive indicators as proposed by Steer provide a clear and well-defined set of monitoring indicators, which relate to aspects which are considered, by some stakeholders, focus areas for RP4. The data is readily available, and the additional effort of involved parties would be limited. However, the indicators have the same drawbacks as the current ones (i.e. not identifying the root causes of degradation of the indicator, changes to

the reporting levels can be driven by variables other than safety performance).

- ⁵⁶ The proposal of EASA provides a view on high (safety) risks defined by the organisations. They also would require little additional effort of involved parties since they make use of the regular work of the ANSPs. They would provide a view of the evolution of the risk picture, including emerging high-risk areas. This valuable information can also be compared and contrasted with that provided in the RP4 Performance Plans.
- Taking into consideration the strengths and weaknesses of the different options, the PRB suggests introducing the indicator proposed by EASA. This would provide a richer picture of safety performance, high-risk areas as identified locally, and the evolution of the risk picture where emerging risks are identified. For monitoring purposes, this is of higher value than the development of a prescriptive set of indicators. The PRB highlights that guidance material is currently being developed by an EASA working group comprised of Member States and Stakeholders.

	Scope				
Indicator	Union- wide	Local	Steer proposal	PRB suggestion	
Number of overloads reported by controller		~	Number of overloads re- ported by the controller.	Do not introduce. Consider EASA proposal.	
ATM-Specific occurrences	~	~	Re-introduce indicator with new definition compared to RP2, potential enhancement of reported data.	Do not introduce. Consider EASA proposal.	
Airspace infringements at all air traffic services units	~	~	Re-introduce indicator with new definition compared to RP2, potential enhancement of reported data.	Do not introduce. Consider EASA proposal.	
Rate of loss of Minimum Safe Altitudes	~	~	Re-introduce indicator, po- tential enhancement of re- ported data.	Do not introduce. Consider EASA proposal.	

La d'Anna ann	Scope		5404	
Indicator	Union- wide	Local	EASA proposal	PRB suggestion
Locally identified indicators in the ANSP Management System		~	Introduce.	Introduce. EASA to define guidelines on the level of information to be provided.

Table 2 – Summary table of new safety indicators proposed for RP4

3 ENVIRONMENT

3.1 Existing indicators within the Regulation

- ⁵⁸ Under Annex I of the Regulation, the following KPI and PIs are used to assess performance within the environment KPA at Union-wide level:
 - The average horizontal en route flight efficiency of the actual trajectory (KEA) (KPI) (Section 1, 2.1);
 - The average horizontal en route flight efficiency of the last filed flight plan (KEP) (Section 1, 2.2(a));
 - The average horizontal en route flight efficiency of the shortest constrained route (SCR) (Section 1, 2.2(b));
 - The effective use of reserved or segregated airspace (Section 1, 2.2(c));
 - The rate of planning via available airspace structures (Section 1, 2.2(d)); and
 - The rate of using via available airspace structures (Section 1, 2.2(e)).
- 59 For the local level, the following indicators are used to assess performance within the environment KPA:
 - The average horizontal en route flight efficiency of the actual trajectory (KPI) (Section 2, 2.1);
 - The average horizontal en route flight efficiency of the last filed flight plan trajectory at local level (Member State) (Section 2, 2.2(a));
 - The average horizontal en route flight efficiency of the shortest constrained trajectory (Section 2, 2.2(b));
 - The additional time in the taxi-out phase (Section 2, 2.2(c));
 - The additional time in terminal airspace (Section 2, 2.2(d));
 - The share of arrivals applying continuous descent operation (Section 2, 2.2(e));
 - The effective use of reserved or segregated local airspace (Section 2, 2.2(f));
 - The rate of planning via available local airspace structures (Section 2, 2.2(g)); and
 - The rate of using available local airspace structures (Section 2, 2.2(h)).

3.2 Review of Steer proposals for existing indicators

60 Steer provides a review of all existing environmental indicators in their report to the Commission. Steer proposes retaining all the existing environment indicators. Table 3 (next pages) summarises the Steer proposals and the PRB suggestions regarding the existing environment indicators.

Average horizontal en route flight efficiency of the actual trajectory (KEA) at Union-wide and local level (Member State)

- 61 Environmental performance is measured through one KPI: Horizontal en route flight efficiency of the actual trajectory (KEA). The indicator measures the additional distance flown of the actual trajectory in addition to the great circle distance. The indicator is defined at Union-wide and local level. For the Union-wide level, the additional distance flown is summed over IFR flights within or traversing the European airspace, while for the local level this is summed over IFR flights within or traversing the local airspace.
- 62 Steer highlights that KEA is not suitable in case of major disturbances (e.g. airspace closure) and is an imperfect proxy that does not capture the impact of the ANSPs' actions. Additionally, the indicator does not account for the impact of weather. Steer highlights that whilst KEA is not an optimal indicator of environmental performance, there is currently no better option and, therefore, Steer suggests retaining the indicator for RP4.
- Given the importance of the environment KPA, the PRB will continue its work to refine the methodologies for measuring horizontal flight efficiency for inclusion as an indicator later in RP4 and for consideration as a KPI in RP5. The PRB also encourages Member States to set local incentive schemes on other appropriate indicators, as specified in articles 10(3) and 11(4) of the Regulation.

Average horizontal en route flight efficiency of the last filed flight plan trajectory (KEP) at Union-wide and local level (Member State)

⁶⁴ The average horizontal en route flight efficiency of the last filed plan trajectory (KEP) is defined as the difference between the length of the en route part of the last filed flight plan trajectory and the corresponding portion of the great circle distance.

- ⁶⁵ Steer suggests retaining this indicator for RP4, as it is an explanatory indicator of the constraining factors that limit horizontal flight efficiency. The indicator helps to understand the environmental performance as measured by KEA. KEP measures the efficiency of the routes planned by airspace users according to their own planning tools and criteria.
- ⁶⁶ The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

Average horizontal en route flight efficiency of the shortest constrained trajectory (SCR) at Union-wide and local level (Member State)

- ⁶⁷ The average horizontal en route flight efficiency of the shortest constrained trajectory (SCR) is defined as the difference between the length of the en route part of the shortest constrained route available for flight planning measured between the exit and entry points of two terminal manoeuvring areas and the corresponding portion of the great circle distance. The indicator is defined at Union-wide and local levels. For the Union-wide level, it is defined over IFR flights within or traversing the European airspace, while for the local level over IFR flights within or traversing the local airspace.
- Steer suggests retaining this indicator. In a similar manner to KEP, it serves as an explanatory indicator of KEA. The relevance of SCR lies in its correlation with traffic volumes, available capacity and airspace availability and restrictions in the flight planning stage. This indicator reflects the options airspace users have when planning their flights to minimise delays.
- 69 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

Additional time in taxi-out phase (AXOT) at local level (airport)

70 The additional time in the taxi-out phase (AXOT) is defined as the difference between the actual taxiout time and the unimpeded taxi-out time at

- Steer suggests retaining this indicator for RP4. Steer highlights that unlike en route indicators, AXOT is generally not impacted by the decisions of other ANSPs and other stakeholders. Hence, this indicator offers a good representation of the actions of each ANSP.
- The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. In addition, the PRB suggests applying the updated methodology of Eurocontrol. The updated methodology relies less on expert judgement, is more easily reproducible, and can be automatically updated on a rolling 12-month basis. The indicator calculated in accordance with the updated methodologies is already available on the ANS Performance platform. Therefore, no additional data collection or calculation are required.²

Additional time in terminal airspace (ASMA) at local level (airport)

- 73 This indicator is defined as the difference between the Arrival Sequencing and Metering Area (ASMA) transit time and the unimpeded time based on ASMA transit times at airports with more than 80,000 annual IFR movements and represents the inefficiencies during the arrival phase at airports. The indicator is defined at local level.
- 54 Steer suggests retaining this indicator for RP4. As for the AXOT indicator, Steer highlights that, unlike en route indicators, terminal areas are less impacted by the decisions of other ANSPs and other stakeholders. Hence, the ASMA indicator offers a good representation of the actions of each ANSP.
- 75 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. In addition, the PRB suggests applying the updated methodology of Eurocontrol. The indicator calculated in accordance with the updated methodologies is already available on the ANS Performance platform. Therefore, no additional data collection or calculation are required.³

airports with more than 80,000 annual IFR movements and represents the inefficiencies during the taxi-out phase at airports.

² Details on updated methodologies and the consultation process can be found <u>here</u>.

³ Details on updated methodologies and the consultation process can be found <u>here</u>.

Share of arrivals applying Continuous Descent Operation (CDO) at local level (airport)

- This indicator is defined as the ratio between the total number of arrivals performing a CDO from a reference point at a height above ground and the total number of arrival operations. The higher the ratio, the higher the flight efficiency during this phase of flight. The indicator is defined at local level and applies to airports with more than 80,000 annual IFR movements.
- 77 Steer suggests retaining this indicator for RP4, as it identifies room for improvement at the individual airports.
- 78 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

Effective use of reserved or segregated airspace (ERSA) at Union-wide and local level

- 79 The indicator is defined as the ratio of the initial requested allocated time for reservation or segregation from general air traffic, and the final allocated time used for the activity requiring such segregation or reservation, as reported to the Network Manager. It measures how well the military or other entity for which an airspace volume has been established are planning and releasing excess airspace. The indicator is defined at Unionwide and local levels.
- Steer suggests retaining this indicator for RP4. Steer highlights that releasing unneeded airspace on the day could have positive contributions to the performance of the network via airspace users flight planning of available routes.
- 81 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. The indicator provides insights into civil-military aspects such as the maturity of the FUA implementation and their impact on environmental performance.

Rate of planning via available airspace structures airspace (RAI) at Union-wide and local level

- ⁸² This indicator is defined as the ratio of aircraft filing flight plans via the airspace structures established by the FUA concept, and the number of aircraft that could have planned through those airspace structures. It shows the rate by which airspace users can plan their flights via available airspace structures to fly the shortest route while considering the airspace the military has released. The indicator is defined at Union-wide and local levels.
- 83 Steer suggests retaining this indicator for RP4. Steer highlights that the application of this indicator is not yet mature given that the data is available for only six Member States.⁴
- The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. The PRB highlights that the indicator is currently calculated and provided by the Network Manager on a monthly basis.

Rate of using via available airspace structures airspace (RAU) at Union-wide and local level

- ⁸⁵ This indicator is defined as the ratio of aircraft flying via the airspace structures established by the FUA concept, and the number of aircraft that could have planned to fly through these airspace structures. It shows the rate by which airspace users use the available airspace structures considering the airspace the military has released. The indicator is defined at Union-wide and local levels.
- Steer suggests retaining this indicator for RP4. As for RAI, Steer highlights that the application of this indicator is not yet mature due to lack of reporting by Member States.
- The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. The PRB highlights that the indicator is currently calculated and provided by the Network Manager on a monthly basis.

⁴ Steer explains that the lack of reporting could be due to the fact that reporting on this indicator requires a lot of effort from Member States.

	Sco	ре		
Indicator	Union- wide	Local	Steer proposal	PRB suggestions
Average horizontal en route flight efficiency of the actual trajectory (KEA) (KPI)	~	~	No change.	No change.
Average horizontal en route flight efficiency of the last filed flight plan trajectory (KEP)	~	~	No change.	No change.
Average horizontal en route flight efficiency of the shortest constrained trajectory (SCR)	~	~	No change.	No change.
Additional time in terminal air- space (ASMA)		~	No change.	Retain. Adopt ECTL up- dated methodology.
Additional time in taxi-out phase (AXOT)		~	No change.	Retain. Adopt ECTL up- dated methodology.
Share of arrivals applying Con- tinuous Descent Operation (CDO)		✓	No change.	No change.
Effective use of reserved or seg- regated airspace (ERSA)	~	~	No change.	No change.
Rate of planning via available airspace structures (RAI)	~	~	No change.	No change.
Rate of using via available air- space structures (RAU)	~	~	No change.	No change.

Table 3 - Summary table of the current environment indicators.

3.3 Review of Steer proposals for new indicators

- 88 In addition to the existing indicators, Steer proposes new environment performance indicators to be introduced in RP4:
 - Horizontal and Vertical TMA; and
- 89 (Estimated) Gate-to-gate fuel burn/Excess fuel burn.Table 4 (next pages) summarises the Steer proposals and the PRB suggestions regarding the new environment indicators for RP4.

Horizontal and Vertical TMA

Steer proposes introducing a new indicator on horizontal and vertical TMA to increase focus on terminal areas (being more directly within the environmental control of ANSPs than en route indicators). This indicator is defined at local level as horizontal and vertical deviations for arrivals from a selected horizon (typically 50NM, up to 200NM) down to final, using two types of reference trajectory (best flown and ideal) to identify airspace and operations related inefficiencies. They measure deviations from reference, "best flown" specific to each airport (operations) and "ideal" common to all airports (airspace). It is a new metric based on Network Manager modelling. Steer highlights that while the indicator is mature, some calibration related to best performer and ideal reference level needs to be done before its introduction. Steer suggests introducing the indicator during the course of RP4 or beyond.

91 The PRB does not suggest including this new indicator for RP4. The PRB accepts that this could be an indicator of aviation's performance but notes that it only focusses on the local aspect of performance, which is already monitored by current indicators Therefore, the PRB suggests maintaining the existing local indicators for RP4 rather than introducing a new horizontal and vertical TMA indicator.

(Estimated) Gate-to-gate fuel burn/Excess fuel burn

Steer proposes introducing a gate-to-gate fuel burn indicator. The scope of this indicator is to measure the total fuel burn and CO₂ emissions of all phases of flight, from taxi-out at the departure airport to taxi-in at the arrival airport. The indicator has the benefit of focussing on both en route

Indiaatan	Scope		Stear propose	
Indicator	Union- wide	Local	steer proposais	PRD suggestions
Horizontal and Vertical TMA		~	Introduce. Some calibration issues need to be solved meaning an intro- duction after the start of RP4.	Do not introduce. Maintain existing indicators.
(Estimated) Gate-to-gate fuel burn/Excess fuel burn	V	V	Introduce. Whilst not all drivers are under ANSP control, useful to have data. Methodology to be finalised, meaning an introduction after the start of RP4.	Introduce only gate-to-gate CO ₂ emis- sions as it covers a representative sample of flights in SES, covers all phases of flights and can be broken down at local level. Whilst it is not fully under the control of the ANSPs, it is an indicator of aviation's CO ₂ per- formance.
En route vertical flight efficiency	~	~	-	Introduce. Complements existing indi- cators. Indicator already existing and monitored by the Network Manager. PRB to work with NM to consider the proposed refinements.
Continuous climb operation (CCO)		~	-	Introduce. Complements existing indi- cators. Indicator already existing and published on the Eurocontrol ANS Performance platform.
Additional taxi-in time (AXIT)		~	-	Introduce. Complements existing indi- cators. Indicator already existing and published on the Eurocontrol ANS Performance platform.

and local aspects and is defined at Union-wide level.

- Steer highlights the benefits of the indicator, 93 which is based on Eurocontrol modelling and does not require actual airspace users' fuel data. Moreover, its reliance on a model makes it is easy to replicate. However, Steer acknowledges that there would be a shared responsibility over this indicator between ANSPs and airspace users as gate-to-gate fuel burn is not fully under ANSP control and it could prove difficult to set a target on fuel burn as the ideal target (or reference) fuel burn level would need to be defined. Steer additionally highlights that the indicator is still under development (e.g. definition of the reference trajectories). Therefore, Steer proposes a delayed introduction of this new indicator during RP4.
- The PRB considers a comparison to a reference 94 would potentially be more beneficial and suggests introducing, in RP4, the additional gate-to-gate CO₂ emissions indicator developed by Eurocontrol. The additional gate-to-gate CO₂ emissions indicator due to ATM and network constraints is defined, at Union-wide level, as the difference between the total CO₂ emissions of the constrained reference trajectory and the total CO₂ emissions of the theoretical reference trajectory.⁵ It is also measured as the additional gate-to-gate CO₂ emissions due to flight planning and execution, and this is calculated as the difference between the total CO₂ emissions of the actual trajectory and the total CO₂ emissions of the constrained reference trajectory. This indicator has the potential to monitor performance at local level as gate-to-gate CO₂ emissions per flight time.
- 95 On a similar note, the PRB considers excess fuel burn (XFB) to be a promising indicator, although it is likely to be affected by many variables, such as Airspace Users' preferences. Developed by the Network Manager, this indicator calculates the excess fuel burn for an airport pair/aircraft type combination based on the total actual fuel burn/total reference fuel burn.⁶ It is currently calculated for flights operated wholly within the

Network Management area, and therefore does not consider flights overflying this airspace or entering/exiting this airspace. It does not require actual fuel burn data from airlines as it is based on Network Manager modelling.

- The PRB acknowledges that gate-to-gate fuel burn 96 and excess fuel burn do not fully reflect the impact of ANSPs; both indicators estimate aviation's fuel burn and CO₂ emission performance across all phases of flight at the Union-wide level. However, the alignment of the performance and charging scheme to the EU's CO₂ reduction goals makes a fuel burn indicator a valuable addition. On this basis, and following the technical meeting with stakeholders on 23rd May 2024, the PRB suggests introducing the gate-to-gate fuel burn indicator for monitoring during RP4. The indicator covers a representative sample of all flights flying within the SES area (including overflights), can be broken down at local level, and measures the gate-to-gate efficiency including the taxi phases. Furthermore, the data used is validated with actual airline data, addressing issues resulting from data estimations.
- 3.4 PRB suggestions for new indicators
- 97 In addition to the indicator proposed by Steer, the PRB suggests the introduction of new performance indicators to extend the coverage of the monitoring of air traffic management performance towards a gate-to-gate approach.
- ⁹⁸ The PRB proposes to include three further environment indicators:⁷
 - En route vertical flight efficiency (Union-wide and local levels);
 - Continuous climb operations (Local level); and
 - Additional taxi-in time (Local level).

En route vertical flight efficiency

99 At present, vertical flight efficiency is not addressed by the performance scheme. This area of environmental flight efficiency represents 16% of

⁵ The constrained reference trajectory represents the CO₂ optimal trajectory taking into account meteorological conditions and ATM and network constraints; while the theoretical reference trajectory represents the CO₂ optimal trajectory taking into account meteorological conditions but no ATM or network constraints. Finally, ATM and network constraints are the impact of the fixed route network, RAD constraints, ATFM regulations, network measures and conditional routes.

⁶ A reference fuel burn is calculated for each airport pair/aircraft type combination, which is the fuel burn of the tenth percentile of all flights in the previous two years for each combination.

⁷ Further to this work, the PRB is considering how measures of flight efficiency could be further improved.

excess CO₂ emissions and is therefore an important measure of performance.⁸

- The Network Manager is currently calculating and monitoring an indicator on vertical flight efficiency measured as the percentage of distance flights spent at optimal flight levels (defined as being at or within 1000ft of the planned flight level).⁹
- 101 The PRB suggests including this indicator at Unionwide and local levels to complement KEA, which only addresses the horizontal efficiency of a flight. However, the PRB suggests refining the indicator to consider:
 - The impact of altitude restrictions on vertical flight efficiency; and
 - That all flights above their planned flight level should be assumed to be optimal.
- 102 This indicator, in its current form, is monitored by the Network Manager, and no additional data collection is required. Refinements to the indicator may require additional data and analysis.

Share of departures applying continuous climb operations (CCO)

- 103 This indicator defines the share of departures applying continuous climb operations. The PRB suggests including this indicator to extend the scope of performance monitoring to more phases of the gate-to-gate flight.
- The indicator is currently measured and reported on the ANS Performance platform of Eurocontrol. The PRB suggests applying the same methodology.¹⁰

Additional taxi-in time (AXIT)

- 105 This indicator defines the average excess time spent during the taxi-in phase, during times that the apron and stands are congested. The PRB suggests including this indicator to extend the scope of performance monitoring to more phases of the flight.
- 106 The indicator is currently measured and reported on the ANS Performance platform of Eurocontrol.¹¹ The PRB suggests applying the same methodology.¹²

⁸ EASA's European Aviation Environmental Report (2022) (PRB elaboration).

⁹ Network Performance Plan 2020-2024, can be found here.

¹⁰ Reported CCO data can be found <u>here</u>.

¹¹ Reported AXIT data can be found <u>here</u>.

¹² The methodology developed by Eurocontrol can be found <u>here</u>.

Indiaatar	Scope		Steen proposile	
Indicator	Union- wide	Local	Steer proposais	PRB suggestions
Horizontal and Vertical TMA		~	Introduce. Some calibration issues need to be solved meaning an intro- duction after the start of RP4.	Do not introduce. Maintain existing indicators.
(Estimated) Gate-to-gate fuel burn/Excess fuel burn	~	V	Introduce. Whilst not all drivers are under ANSP control, useful to have data. Methodology to be finalised, meaning an introduction after the start of RP4.	Introduce only gate-to-gate CO ₂ emis- sions as it covers a representative sample of flights in SES, covers all phases of flights and can be broken down at local level. Whilst it is not fully under the control of the ANSPs, it is an indicator of aviation's CO ₂ per- formance.
En route vertical flight efficiency	~	~	-	Introduce. Complements existing indi- cators. Indicator already existing and monitored by the Network Manager. PRB to work with NM to consider the proposed refinements.
Continuous climb operation (CCO)		~	-	Introduce. Complements existing indi- cators. Indicator already existing and published on the Eurocontrol ANS Performance platform.
Additional taxi- in time (AXIT)		~	-	Introduce. Complements existing indi- cators. Indicator already existing and published on the Eurocontrol ANS Performance platform.

Table 4 - Summary table of the new environment indicators proposed for RP4.

4 CAPACITY

4.1 Existing indicators within the Regulation

- 107 Under Annex I of the Regulation, the following KPI and PIs are used to assess the Union-wide performance within the capacity KPA:
 - The average minutes of en route ATFM delay per flight attributable to air navigation services (KPI) (Section 1, 3.1);
 - The average time, expressed in minutes, of arrival ATFM delay per flight attributable to terminal and airport air navigation services (Section 1, 3.2(a));
 - The percentage of flights with en route ATFM delay greater than 15 minutes (Section 1, 3.2 (b)); and
 - The average time, expressed in minutes, of allcause departure delay per flight (Section 1, 3.2 (c)).
- 108 For the local level, the following KPI and PIs are used to assess the performance within the capacity KPA:
 - The average minutes of en route ATFM delay per flight attributable to air navigation services (KPI) (Section 2, 3.1(a));
 - The average time, expressed in minutes, of arrival ATFM delay per flight attributable to terminal and airport air navigation services (KPI) (Section 2, 3.1(b));
 - The percentage of flights adhering to their ATFM departure slots at local level (Section 2, 3.2(a));
 - The average minutes of air traffic control predeparture delay per flight caused by take-off restrictions at the departure airport (Section2, 3.2(b)); and
 - The average time, expressed in minutes, of departure delay from all causes per flight, calculated at local level (Section2, 3.2(c)).

4.2 Review of Steer proposals for existing indicators

109 Steer reviewed all existing capacity performance indicators. Table 5 (next pages) summarises the Steer proposals and PRB suggestions regarding the existing capacity indicators.

Average minutes of en route ATFM delay per flight attributable to air navigation services at Union-wide and local levels

- 110 The average minutes of en route ATFM delay is the only capacity KPI at Union-wide level. It is also defined and calculated as a local level KPI.
- 111 Steer suggests retaining this indicator at Unionwide and local levels since the indicator is welldocumented, and the data collection and the calculation is automated and not subject to human interpretation. Steer highlights several shortcomings of this indicator: (i) ATFM delays are attributed only to one single air navigation service provider for each flight, and only to one delay reason; (ii) the indicator fails to differentiate between situations in which an ANSP is not able to deliver more capacity than planned from situations when the ANSP fails to deliver the planned capacity; and (iii) the indicator covers causes which are outside the control of the ANSPs and it cannot easily be translated into passenger experience, given the buffers in the schedules of airlines.
- 112 The PRB suggests complementing this indicator with a new indicator measuring the distribution of ATFM delays (as defined below in the new indicators for RP4).

Average arrival ATFM delay per flight attributable to airport and terminal air navigation services at Unionwide and local levels

- 113 This indicator is defined as the average arrival ATFM delay per inbound IFR flight. It is calculated at Union-wide level, and also defined as a KPI at local level.
- While Steer suggests retaining this indicator as Union-wide and local levels, it highlights the same shortcomings as for the ATFM delay per flight indicator (both indicators cover ATFM delays with the differences being the geographical scope and the service provider behind the delay). Specifically for arrival ATFM delays, Steer highlights that arrival ATFM delays are more useful when measuring performance than departure delays, as they are more closely related to the punctuality of operations experienced by airspace users and passengers.
- 115 As for the previous indicator, the PRB suggests complementing this indicator with a new indicator

measuring the distribution of ATFM delays (as defined below in the new indicators for RP4).

Percentage of flights with en route ATFM delay greater than 15 minutes at Union-wide level

- 116 This indicator is calculated as the percentage of fights with ATFM delay greater than 15 minutes and covers all IFR flights and all ATFM delay causes, excluding exceptional events. It is defined at Union-wide level.
- Steers suggests removing this indicator for RP4. While it measures an important aspect of operational service quality as it indicates the frequency of relatively longer ATFM delays experienced by airspace users, the introduction of a potential new performance indicator on the average en route ATFM delay per delayed flight would make this indicator redundant. On this basis, Steer suggest removing the indicator to alleviate the administrative burden on stakeholders.
- 118 The PRB agrees with Steer's proposal and suggests removing this indicator for RP4, provided that the indicator on the distribution of ATFM delays is introduced (as defined below in the new indicators for RP4). While the information provided by this indicator is important due to its link with the rerouting of flights, more granular information is required to improve monitoring.

Average time of all-cause departure delay per flight at Union-wide and local levels

- 119 This indicator is defined as the average delay attributable to airline operation, en route ATFM delay reported by airspace users, reactionary delay, and airport operations delays. It is defined at Union-wide and local levels.
- 120 Steer suggests retaining this indicator for RP4 and highlights that an indicator on all-cause departure delays provides a proxy of the disturbances to airline schedules. This provides an important view of operational performance. This indicator can also

be used to indicate the resilience of the network to day-to-day perturbations.

121 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

The percentage of flights adhering to their ATFM departure slots at local level

- 122 This indicator is defined as the average time, expressed in minutes, of departure delay from all causes per flight. It is defined at local level.
- 123 Steer suggests retaining this indicator for RP4, highlighting that the Network Manager uses it to assess the predictability of the network.
- 124 The PRB suggests removing this indicator for RP4. The time window of ATFM slots is 15 minutes (from 5 minutes before to 10 minutes after the assigned calculated take-off time). THE PRB considers this to be too wide, when compared to the average flight time in Europe, to indicate punctuality from an airline perspective.

Average minutes of ATC pre-departure delay per flight caused by take-off restrictions at the departure airport at local level

- 125 This indicator is defined as the average air traffic control pre-departure delay per outbound IFR flight and includes all IFR flights taking off at the departure airport covering delays in start-up caused by air traffic control constraints when the aircraft is ready to leave the departure stand. It is defined at local level.
- 126 Steer suggests retaining this indicator for RP4. Steer notes that this indicator focuses on pre-departure delays rather than arrival delays, and as such, may be more detached from the punctuality experienced by the users. At the same time, the indicator helps to understand the operational performance at the airport level and complements the indicator on airport arrival ATFM delays.
- 127 The PRB agrees with Steer's proposal and suggests retaining the indicator for RP4.

Indicator	Scope		Stoor proposal	
Indicator	Union- wide	Local		PRD suggestions
The average minutes of en route ATFM delay per flight at- tributable to air navigation ser- vices (KPI)	~	~	No change.	No change. Complemented by indicator on the distribution of ATFM delays.
The average time, expressed in minutes, of arrival ATFM delay per flight attributable to termi- nal and airport air navigation services (KPI)		V	No change.	No change. Complemented by indicator on the distribution of ATFM delays.
The average time, expressed in minutes, of arrival ATFM delay per flight attributable to termi- nal and airport air navigation services	~		No change.	No change. Complemented by indicator on the distribution of ATFM delays.
The percentage of flights with en route ATFM delay greater than 15 minutes	~		Remove. Redundant with the introduction of indicator on Minutes of ATFM delay per delayed flight.	Remove if an indicator on the distribution of ATFM delays is introduced.
The average time, expressed in minutes, of all-cause departure delay per flight	~	~	No change.	No change.
The percentage of flights adher- ing to their ATFM departure slots at local level		~	No change.	Remove. The 15-minute width of the slot is too wide to be meaningful for flights with a duration of 60-150 minutes.
The average time, expressed in minutes, of departure delay from all causes per flight		~	No change.	No change.

Table 5 - Summary table of the current capacity indicators.

4.3 Review of Steer proposals for new indicators

- 128 In addition to the existing indicators, Steer proposes new capacity performance indicators to be introduced in RP4:
 - Minutes of ATFM delay per delayed flight (Union-wide and local levels);
 - Weighted ATFM delays (Union-wide and local levels);
 - Days where throughput is above expected traffic and where delays per flight are below reference value (Local level); and
 - On time in full, OTIF (Union-wide and local levels).
- 129 Table 6 (next pages) summarises the Steer proposals and the PRB suggestions regarding the new capacity indicators for RP4.

Minutes of ATFM delay per delayed flight

- 130 Steer suggests introducing the minutes of ATFM delay per delayed flight as a Union-wide and local level indicator.
- 131 This is a variant of the existing capacity KPI, where the average en route ATFM delay is only calculated for flights which are subject to en route ATFM delay. Steer highlights that this indicator could potentially replace the indicator on the percentage of flights with en route ATFM delays greater than 15 minutes since it provides a more comprehensive picture of en route ATFM delays. Data for this indicator is already available in the existing datasets used for the calculation of ATFM delays.
- 132 The PRB agrees with Steer's proposal and suggests introducing this new capacity indicator for RP4.

Weighted average ATFM delays per flight

- 133 Steer suggests introducing another variant of the average ATFM delay indicator, where the average delay per flight is calculated based on a set of preliminarily agreed weights. The weights would put more emphasis on delays occurring during critical operational periods (e.g. during first rotation hours or peak demand periods within the day). This indicator would be calculated on the Unionwide and local level.
- 134 It would provide additional information on performance aspects that are more important to airspace users and thus could bring value to measuring capacity performance.

- 135 Steer does not set out how the weights should be calculated and defined, noting that preliminary work on the calibration and the agreement from stakeholders is necessary.
- 136 In terms of data availability, this indicator would largely use the same datasets as all other indicators on ATFM delays, with the only difference being the weighting system.
- 137 Whilst the PRB agrees that there could be value in focusing on delays produced at critical times of the days, the PRB suggests that the definition of a suitable weighting system for both Union-wide and local levels would be difficult. Furthermore, indicators focussing on the length of delays and/or on the capacity provided by ANSPs are more important when monitoring performance. On balance, and taking into consideration the administrative burden placed on stakeholders, the PRB does not suggest the inclusion of this new indicator for RP4.

Days where throughput is above expected traffic and where delays per flight are below reference value

- 138 Steer suggests introducing this indicator to capture the capacity provided by ANSPs and to reconcile this information with traffic demand and ATFM delays. Steer proposes to introduce the indicator at local level.
- 139 The indicator identifies capacity provision by measuring airspace throughput over a day in terms of IFR movements and comparing it to the forecasted traffic demand. Combining this with the information on when ATFM delays were below the local targets/breakdown values, the indicator would show when ANSPs were able to accommodate unexpected traffic growth without generating excess delays.
- 140 Steer suggests that if this indicator is not found suitable during the testing period in early RP4, an indicator on "three-hour throughput" or sectoropening hours should be introduced instead. The PRB understands the operational importance of measuring the sustained peak-hour throughput of ACCs, as this indicates an important aspect of the achieved capacity. However, there are limitations in aggregating a throughput indicator, and the absence of a mature methodology to of comparing the monitored figures to an appropriate reference figure. Given these limitations, the PRB suggests

that the concept should be further explored during RP4 and potentially introduced in RP5.

- 141 The indicator has the benefit of connecting information on traffic demand, airspace throughput, and ATFM delays. It also has a strong connection to the capacity planning collaborative decision making process run by the Network Manager together with the ANSPs. However, the PRB highlights that this indicator cannot be aggregated from the level of ACCs to ANSPs, and beyond (i.e. it is not additive).
- 142 To facilitate the interpretation of the indicator, the PRB suggests defining the indicator as the percentage of en route ATFM delay minutes that occurred on days when the daily throughput of the ACC was above the daily traffic forecast, compared to the total en route ATFM delay minutes generated over the calendar year. The indicator would then show when ANSPs were unable to manage unforeseen traffic increases and when they were unable to provide sufficient capacity to accommodate expected traffic demand.
- 143 The PRB also suggests introducing an indicator on the sector-opening hours (as defined in the next sections) to complement the indicator by capturing other aspects of capacity provision.

On time in full (OTIF)

- 144 Steer suggests introducing this indicator to capture a comprehensive view of operational performance. It consists of two components: The efficiency of the trajectory of the flights, and delays experienced by the flights. Steer proposes to define this indicator at Union-wide and local levels.
- 145 The indicator is calculated as the percentage of flights able to fly their optimal trajectory without ATFM delays. Steer notes that it is still under development, primarily to define what "optimal" trajectory would mean. Steer suggests that the trajectory included in the flight plan could be used as a starting point until a better definition can be established. Steer proposes introducing the indicator at Union-wide and local levels. Data is available for this indicator, as the information is the same as for all other indicators of ATFM delays and environmental performance.
- 146 The PRB highlights that this indicator combines environment and capacity performance aspects, capturing interdependencies between the two KPAs. Therefore, the PRB agrees with Steer's

proposal and suggests introducing this new indicator for RP4. The PRB suggests using the gate-togate excess CO_2 emission indicator as a reference for the efficiency of the flight trajectories. This eliminates the need to define the notion of optimal trajectory linking the indicator with one of the new environment performance indicators suggested.

- 147 The indicator can be calculated by counting the number of IFR flights within an airspace block which satisfy the two following criteria:
 - The flight was not subject to en route ATFM delay; and
 - The flight did not generate excess CO₂ emissions while flying in the SES area.

The number of flights meeting both criteria would then be divided by the total number of IFR flights in the airspace block and multiplied by 100 to arrive to a percentage figure.

4.4 PRB suggestions for additional new indicators

- 148 The PRB suggests the introduction of two additional capacity indicators to enrich the monitoring of capacity performance and to focus on capacity provision:
 - Distribution of IFR flights per length of en route and airport arrival ATFM delay (Unionwide and local levels); and
 - Sum of sector-opening hours (Union-wide and local levels).
- 149 These new indicators address the shortcomings of the existing capacity KPIs by providing insights on the operational reality represented by the average ATFM delay per flight.

Distribution of IFR flights per length of en route and airport arrival ATFM delays

- 150 This indicator identifies the percentage of IFR flights (or IFR arrivals for airport arrival ATFM delays) which were subject to ATFM delays of different lengths. It should be calculated for en route and terminal and based on the following delay categories:
 - Zero ATFM delay;
 - ATFM delays not greater than 5 minutes;
 - ATFM delays greater than 5 but not greater than 15 minutes;

- ATFM delays greater than 15 but not greater than 30 minutes;
- ATFM delays greater than 30 but not greater than 60 minutes; and
- ATFM delays greater than 60 minutes.

The indicator should be defined at Union-wide and local levels, and for en route and arrival ATFM delays. For each category, the percentage value is calculated by counting the number of flights/arrivals subject to the relevant amount of ATFM delay and dividing that figure by the total number of flights that have been subject to ATFM delays (i.e. delayed flights/arrivals).

- 151 The category with zero ATFM delay includes flights which were not subject to ATFM delays. The second category represents flights which experience some ATFM delays, however, the duration of the delay was rather short, likely without any significant operational impact. The category with ATFM delay between 5 and 15 minutes shows flights which were subject to ATFM delays that may result in schedule disruptions especially if the delays occurred early in the day. The category with ATFM delays between 15 and 30 minutes represents situations where airlines are likely to re-plan their flights and reroute around congested airspace, thus generating environmental impacts. The categories with ATFM delays greater than 30 minutes and 60 minutes represent situations of severe disruptions locally and, in the case of the last category (greater than 60 minutes), most likely on the network level as well.
- 152 The above categories are used for analysing both en route ATFM delays and airport arrival ATFM delays. IFR arrivals are considered for airport arrival ATFM delays.
- 153 The PRB suggests introducing the indicator as it would provide a more comprehensive picture of ATFM delays and would replace the existing indicator on the annual percentage of flights with ATFM delays greater than 15 minutes. The PRB currently monitors this distribution for en route ATFM delays as presented in the annual monitoring reports since 2021.
- 154 Data is available for this indicator, as ATFM delays are already measured for each flight. This means that no additional data collection is required for this indicator.

The sum of sector-opening hours

- 155 This indicator identifies the capacity that ANSPs are able to provide during a particular period. It complements the information provided by the ATFM delay indicators. The PRB considers that this indicator focuses on the ability of ANSPs to provide capacity, without distortions presented by traffic figures and delays. As sector-opening hours are closely related to the allocation and use of the resources of the ANSPs, this indicator would place a focus on the interdependency between the capacity and cost-efficiency KPAs (more details in the cost-efficiency section).
- 156 The sum of sector-opening hours is defined as the total opening time of each ATS sector that was activated over a calendar year. For example, the opening time is measured as follow:
 - One sector open for one hour is measured as one sector-opening hour;
 - Two sectors open for one hour is measured as two sector-opening hours;
 - Two sectors open for 30 minutes (i.e. half an hour) equals one sector-opening hour.

The total number of sector-opening hours can be calculated as the product of the number of sectors in each sector configuration and the time during which each configuration was active. Sector-opening hours are additive, therefore values from different ATS units or different periods can be aggregated. This indicator should be defined at both Union-wide and local levels, covering all the ATS sectors which were activated for providing en route ATS services.

- 157 The PRB suggests that this indicator would provide an initial evaluation of the links between resources used and capacity provided by the ANSPs. The PRB has monitored this indicator, as presented, in its annual monitoring reports since 2021. This indicator is a measure for capacity performance when monitored in conjunction with the other capacity indicators.
- 158 ANSPs provide data to the Network Manager on sector numbers and sector-opening hours; this indicator can be accurately calculated from that dataset.

Indicator	Indicator		Steer proposel		
	Union- wide	Local	steer proposal	PRB suggestions	
Minutes of ATFM delay per delayed flight	~	~	Introduce. It can be generated by the Network Manager without concerns.	Introduce. Complemented by indi- cator on delay distribution	
Weighted ATFM delays	~	~	Introduce. The choice of the weights may need some consider- ation but could be done for the start of RP4.	Do not introduce. Better to have indicator on delay distribution.	
Days where throughput is above expected traffic and where delays per flight are below refer- ence value		V	Introduce if proven to work. How- ever, it may require some engage- ment with stakeholders which may mean a delayed introduction of this indicator after the start of RP4. If this indicator is not found to work during RP4, consider in- troducing "3h throughput" or "number of sector hours pro- vided" as alternatives.	Introduce as the percentage of en route ATFM delay minutes that occurred on days when the daily through-put of the ACC was above the daily traffic forecast, com- pared to the total en route ATFM delay minutes generated over the calendar year. Complemented by the sum of sector-opening hours provided.	
On time in full, OTIF	~		Introduce. The issue of the defini- tion of the "optimal" trajectory (whether based on flight plan or request on the day) as well as data generation may mean a de- layed introduction of this new in- dicator after the start of RP4	Introduce. Combines environment and capacity performances. Moreover, it links performance with actual service quality levels.	
Distribution of IFR flights per length of en route ATFM delay	√	~	-	Introduce. Provides a comprehen- sive view of the operational reality behind average delays per flight. Included in PRB Monitoring Re- ports since 2021.	
Distribution of IFR arrivals per length of airport arrival ATFM delay	~	~	-	Introduce. Provides a comprehen- sive view of the operational reality behind average delays per flight.	
Sum of sector- opening hours	~	~	-	Introduce. Important to under- stand performance behind delays. Included in PRB Monitoring Re- ports since 2021	

 Table 6 – Summary table of the new capacity indicators proposed for RP4.

5.1 Existing indicators within the Regulation

- 159 Under the Regulation, the following KPI and PIs are used to assess the performance of the costefficiency KPA at Union-wide level:
 - The year-on-year change of the Union-wide determined unit cost (DUC) (KPI) (Section 1, 4.1); and
 - The actual unit cost incurred by users separately for en route and terminal air navigation services at Union level (Section 1, 4.2).
- 160 For the local level, the following KPI and PIs are used to assess the performance of the cost-efficiency KPA:
 - The DUC for en route and terminal air navigation services (KPI) (Section 2, 4.1(a) and 4.1(b)); and
 - The actual unit cost incurred by users separately for en route and terminal air navigation services (Section 2, 4.2)

5.2 Review of Steer proposals for existing indicators

161 Steer provides a review of all the existing cost-efficiency performance indicators, suggesting no changes or modifications. Table 7 (next page) summarises the Steer proposals and the PRB suggestions regarding the existing cost-efficiency indicators.

The year-on-year change of the Union-wide determined unit cost (DUC)

- 162 This indicator is defined as the percentage yearon-year change of the Union-wide determined unit cost, and it is the only KPI at the Union-wide level. This indicator is defined for en route activities.
- 163 Steer suggests retaining this indicator as Unionwide level since the indicator is widely accepted

by stakeholders and has influenced outturn performance over previous regulatory periods. Steer notes that the main issue of this indicator is that it does not allow for a full understanding of the interrelationship between cost and capacity provided.

164 The PRB suggests accompanying this indicator with more information on the cost for capacity provided (as defined below in the new indicators for RP4).

The actual unit cost incurred by users separately for en route and terminal air navigation services at Union-wide and local levels

- 165 This indicator is defined as the sum of the determined unit cost and the adjustments stemming from the specific year. It is defined at Union-wide and local levels, and for en route and terminal activities.
- 166 Steer suggests retaining this indicator, as it provides complementary information to the cost-efficiency KPIs.
- 167 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.

The DUC for en route and terminal air navigation services at local level

- 168 This indicator is defined as the yearly determined unit cost calculated as the ratio between the determined costs and the forecast traffic as defined in the performance plan. It is defined as the local level KPI, for en route and terminal activities.
- 169 Steer suggests retaining this indicator, highlighting similar shortcomings as of the Union-wide KPI.
- 170 As for the Union-wide indicator, the PRB suggests accompanying this indicator with more information on the cost for capacity provided (as defined below in the new indicators for RP4).

Indicator	Scope		Steer proposal	DDB suggestions	
	Union- wide	Local			
Year-on-Year change of Un- ion-wide DUC (KPI)	~		No change.	No change. Complemented by indicator on cost of capacity provided.	
DUC at charging zone for en route and terminal ANS (KPI)		~	No change.	No change. Complemented by indicator on cost of capacity provided.	
Actual unit cost incurred by users separately for en route and terminal ANS	~	~	No change.	No change.	

Table 7 – Summary table of the current cost-efficiency indicators.

5.3 Review of Steer proposals for new indicators

- 171 In addition to the existing indicators, Steer proposes new cost-efficiency performance indicators to be introduced in RP4:
 - Total economic cost (Union-wide and local levels).
- 172 Table 8 summarises the Steer proposals and the PRB suggestions regarding the cost-efficiency indicators for RP4.

Total economic cost

- 173 This indicator identifies the cost of service provision and the cost of delay generated by ANSPs. Steer suggests that it provides an economic translation of the "true" costs borne by airspace users in relation to the provision of ATM. Steer suggests introducing this indicator while acknowledging that the cost of delay (included in the calculation) is based on a study last updated in 2015. Steer also proposes that, to ensure implementation from the start of RP4, the cost of emissions is not included in the calculation of total economic cost.
- 174 Whilst the PRB agrees that the indicator brings value by potentially connecting information from three key performance areas, the PRB has some concerns that it would be based on dated cost of delay estimates leading to, potentially, misleading results. The PRB considers that this indicator may

usefully be introduced once cost of delay estimates are updated including all relevant items.

- 5.4 PRB suggestions for additional new indicators
- 175 The PRB suggests the introduction of one cost-efficiency indicator:
 - Cost per unit of capacity (Union-wide and local levels).

Cost per unit of capacity

- 176 This indicator would identify the relation between costs and provision of capacity and thereby place a focus on the costs that ANSPs face to provide capacity. This would provide some clarity to this aspect of the performance which would evolve over time. As highlighted in the capacity section, this indicator complements the sector-opening hours indicator proposed.
- 177 The indicator would be calculated as the ratio of en route actual costs and the sum of sector-opening hours provided by the area control centres of each ANSP and defined at Union-wide and local levels. As for the capacity indicator, the PRB acknowledges that this indicator would not be suitable as a standalone measure and should be considered in conjunction with the other capacity indicators.

	Sco	ре		
Indicator	Union- wide	Local	Steer proposal	PRB suggestions
Total economic cost	V	V	Introduce. This indicator ad- dresses interdependencies be- tween capacity and cost-effi- ciency KPAs. Do not include envi- ronment KPA as part of total eco- nomic cost in this indicator in RP4 as it will not be ready yet.	Do not introduce, until the cost of delay is updated (current study is dated).
Cost per unit of capacity	~	~	-	Introduce. Costs per sum of sector- opening hours facilitates an under- standing of interdependency be- tween capacity and cost and would complement the cost-efficiency KPI.

Table 8 - Summary table of the new cost-efficiency indicators proposed for RP4.

6 NETWORK MANAGER

6.1 Safety indicators

Existing indicators

- 178 Under Annex I of the Regulation, the following KPI and PI are used to assess the performance of the safety performance area of the network functions:
 - Minimum level of the Effectiveness of Safety Management (EoSM) (KPI) (Section 3, 2.1); and
 - Percentage of ATFM over-deliveries (Section 3, 2.2).

Proposals for existing and new indicators

- 179 For the safety KPA, Steer suggests retaining the current indicators. In addition, Steer suggest updating the EoSM questionnaire for the Network Manager. The PRB agrees with Steer, highlighting that the RP4 EoSM questionnaire for the Network Manager has been developed.
- 180 EASA proposes a new indicator to complement the ones proposed at Union-wide and local levels. This indicator covers the locally identified indicators in the Network Manager management system having the highest risk at network level.
- 181 The PRB agrees with EASA and suggests introducing the indicator as this would complement the indicators suggested at the Union-wide, Member State, and ANSP level. The indicator would provide the Network Manager view of the network level risk picture and the evolution of the picture over time. This indicator is consistent with the one defined at the ANSP level.

6.2 Environment indicators

Existing indicators

- 182 Under Annex I of the Regulation, the following KPI is used to assess the performance of the environment performance area of the network functions:
 - The en route flight efficiency improvement generated by the European Route Network Design function related to the last filed plan trajectory (KPI) (Section 3, 3.1).

Proposals for existing and new indicators

- 183 Steer suggests retaining the current indicator without any change and to not introduce additional indicators.
- 184 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4. However, the PRB suggests improving the current indicators during RP4 while preparing potential future indicators (e.g. on Airspace and ARES availability).
- 185 Moreover, the PRB suggests including the route extensions resulting from the actions of the Network Manager Operations Centre (NMOC) to reroute traffic to reduce delay. The effect of these actions demonstrates the interdependency between the capacity and environment KPAs and should be monitored.

6.3 Capacity indicators

Existing indicators

- 186 Under Annex I of the Regulation, the following KPIs and PIs are used to assess the performance of the network functions in the capacity key performance area:
 - Percentage of en route ATFM delay savings from the CDM network procedures and Network Manager Operations Centre actions over the total year-on-year en route ATFM delay savings (KPI) (Section 3, 4.1(a));
 - Percentage of arrival ATFM delay savings from the CDM network procedures and Network Manager Operations Centre actions, over the total arrival ATFM delay savings (KPI) (Section 3, 4.1(b));
 - Annual percentage of IFR flights with ATFM delay above 15 minutes (Section 3, 4.2(a));
 - The average of the daily number of ATFM regulations that each produces less than 200 minutes of delay (Section 3, 4.2(b));
 - The average of en route ATFM weekend delay expressed in minutes of delay per flight (Section 3, 4.2(c));
 - The annual percentage of first rotation delay due to capacity and staffing for a pre-selection of area control centres/airports with the most significant potential delay reduction as identified annually by the Network Manager (Section 3, 4.2(d));

- The effective use of reserved or segregated airspace (ERSA) (Section 3, 4.2(e));
- The rate of planning via available airspace structures (RAI) (Section 3, 4.2(f)); and
- The rate of using available airspace structures (RAU) (Section 3, 4.2(g)).

Proposals for existing and new indicators

- 187 As regards the two KPIs on en route and airport arrival ATFM delay savings, Steer highlights that the regulatory definition is difficult to interpret and calculate. Given these issues, the Network Manager uses an alternative interpretation and calculation methodology: Measuring delay savings stemming from CDM network procedures and NM Operations Centre actions against a "do-nothing" scenario.
- 188 Steer suggests retaining these KPIs while refining their definitions to be aligned with the existing practice.
- The PRB agrees with Steer's comments, noting the discrepancy between the regulatory definition of the KPIs and the existing calculation methodology. However, the PRB suggests that the indicators are defined in a more transparent way to allow for better performance measurement instead of aligning the regulatory definition with the current practice.
- 190 The PRB suggests that the capacity KPIs for the network functions are defined as the percentage of total minutes of ATFM delays saved as a result of CDM network procedures and NM Operations Centre actions over the original total minutes of ATFM delays, where the saved delay is calculated as the difference between the original delay an IFR flight would have been subject to without the measures and the actual en route ATFM delay of the flight. This approach should be valid for both en route and airport arrival KPIs.
- 191 This approach would ensure that the counterfactual for measuring delay savings is well-defined and would reflect the performance of the network manager in mitigating ATFM delays better.
- 192 Data for using this methodology is available at the Network Manager, given its continuous monitoring of ATFM regulations and ATFM delays.
- 193 Regarding the performance indicators (PIs):
 - Steer suggests retaining the indicator on the percentage of flights with ATFM delays

greater than 15 minutes as it provides important insight into the operational reality.

- Steer suggests retaining the indicator on ATFM regulations generating less than 200 minutes, despite the level of the threshold for the indicator being unclear;
- Steer suggests retaining the indicator on the average weekend ATFM delay per flight.
- Steer suggests retaining the indicator on the first rotation capacity and staffing-related ATFM delays. and
- Steer suggests eliminating the duplication of the three airspace-related indicators (i.e. ERSA, RAI, RAU) between the environment KPA and network manager sections.
- 194 Having considered Steer's proposals, the PRB suggests:
 - Removing the indicator on ATFM delays greater than 15 minutes. This information is already monitored at Union-wide level and would still be available as part of the indicator on the distribution of IFR flights based on the experienced length of ATFM delays.
 - Removing the indicator on ATFM regulations generating less than 200 minutes of delay, as there are no clear benefits identified from the monitoring of this indicator.
 - Retaining the indicator of the weekend ATFM delays if the weighted average ATFM delay indicator as a Union-wide level capacity indicator is not introduced (as suggested by the PRB). However, if the weighted average ATFM delays indicator is introduced at Union-wide level, this indicator might become redundant and should be removed.
 - Retaining the indicator on the first rotation delays while extending its scope to cover all-cause first rotation delays and defining the duration of first rotation.
 - Eliminating the duplication of the three airspace-related indicators (i.e. ERSA, RAI, RAU) in the network functions, while retaining them within the environment KPA.
- 195 No additional indicators are proposed by the PRB or Steer.
- 196 Table 9 (next page) summarises the Steer proposals and the PRB suggestions regarding the capacity indicators for the network functions.

6.4 Cost-efficiency indicators

Existing indicators

- 197 Under Annex I of the Regulation, the following indicator is used to monitor the performance of the cost-efficiency performance area of the network functions:
 - The unit cost for the execution of the tasks of the Network Manager (Section 3, 5.1).

Proposals for existing and new indicators

- 198 Steer suggests retaining the current indicator without any change.
- 199 The PRB agrees with Steer's proposal and suggests retaining this indicator for RP4.
- 200 No additional indicators are proposed by the PRB nor Steer.

Indicator	Steer proposal	PRB proposal
Percentage of ATFM delay savings from the CDM net- work procedures and NM Operations Centre actions (en route and arrival ATFM delays) (KPIs)	Change definition to the one currently in use.	Change. Definition in the regulation is impossible to apply. Savings to be calcu- lated as the difference be- tween actual ATFM delay and the original delay (i.e. the one the flight would have been subject to based on the network situation).
Percentage of flights with greater than 15 minutes of en route ATFM delay	No change.	Remove, it is already in- cluded in capacity KPA.
Average daily number of ATFM regulations producing max 200 minutes of delay	No change.	Remove, no real benefit of monitoring identified.
Average en route ATFM weekend delay per flight	No change.	No change. Remove only If weighted average delay indi- cator is introduced.
Annual percentage of first rotation delay due to capacity and staffing	No change.	Change to measure all-cause first rotation delays and de- fine first rotation period.
ERSA, RAI, RAU	No change. To consider the duplications.	Remove, duplicate with envi- ronment KPA.

Table 9 – Summary table of the capacity indicators of network functions.

7 CONCLUSIONS

201 The following tables summarise the final indicators that the PRB suggests removing, retaining, revising, or introducing in RP4.

Safety

	Scope		
Indicator	Union- wide	Local	PRB suggestions
Minimum level of the Effectiveness of Safety Management (EoSM) (KPI)	~	~	No change, revise EoSM.
The rate of runway incursions at Union-wide level with a safety impact	~		No change. Grouping can be done based on al- ready reported data.
The rate of separation minima infringements at Union-wide level with a safety impact	~		No change.
The rate of runway incursions at local level (Member State)		~	No change. Grouping can be done based on al- ready reported data.
The rate of separation minima infringements within the airspace of all controlling air traffic services units at local level (Member State)		~	No change.
The rate of runway incursions with ATS/CNS contribution at local level (airport)		~	No change. Grouping can be done based on al- ready reported data.
The rate of separation minima infringements with ATS/CNS contribution at local level (ANSP)		~	No change.
Application by the ANSPs of automated safety data recording systems		~	Remove. The indicator has little value for safety performance monitoring.
Locally identified indicators in the ANSP Man- agement System		~	Introduce. EASA to define guidelines on the level of infor- mation to be provided.

Environment

	Scope		
Indicator	Union- wide	Local	PRB suggestions
Average horizontal en route flight efficiency of the actual trajectory (KEA) (KPI)	~	~	No change.
Average horizontal en route flight efficiency of the last filed flight plan trajectory (KEP)	~	~	No change.
Average horizontal en route flight efficiency of the shortest constrained trajectory (SCR)	~	~	No change.
Additional time in terminal airspace (ASMA)		✓	No change. Adopt ECTL updated methodology.
Additional time in taxi-out phase (AXOT)		✓	No change. Adopt ECTL updated methodology.
Share of arrivals applying Continuous Descent Operation (CDO)		~	No change.
Effective use of reserved or segregated air- space (ERSA)	~	~	No change.
Rate of planning via available airspace struc- tures (RAI)	~	~	No change.
Rate of using via available airspace structures (RAU)	~	~	No change.
(Estimated) Gate-to-gate fuel burn/Excess fuel burn	~	V	Introduce the gate-to-gate CO2 emissions as it covers a representative sample of flights in SES, covers all phases of flights and can be broken down at local level. Whilst it is not fully under the control of the ANSPs, it is an indicator of avia- tion's CO2 performance.
En route vertical flight efficiency	~	~	Introduce. Complements existing indicators. Indi- cator already existing and monitored by the Net- work Manager. PRB to work with NM to consider the proposed refinements.
Continuous Climb Operation (CCO)		~	Introduce. Complements existing indicators. Indi- cator already existing and published on the Euro- control ANS Performance platform.
Additional taxi-in time (AXIT)		~	Introduce. Complements existing indicators. Indi- cator already existing and published on the Euro- control ANS Performance platform.

Capacity

	Scope			
Indicator	Union- wide	Local	PRB suggestions	
The average minutes of en route ATFM delay per flight attributable to air navigation ser- vices (KPI)	~	~	No change. To complement with indicator on the distribution of ATFM delays.	
The average time, expressed in minutes, of ar- rival ATFM delay per flight attributable to ter- minal and airport air navigation services (KPI)		~	No change. To complement with indicator on the distribution of ATFM delays.	
The average time, expressed in minutes, of ar- rival ATFM delay per flight attributable to ter- minal and airport air navigation services	~		No change. To complement with indicator on the distribution of ATFM delays.	
The percentage of flights with en route ATFM delay greater than 15 minutes	~		Remove only if indicator on the distribution of ATFM delays is introduced.	
The average time, expressed in minutes, of all- cause departure delay per flight	~	~	No change.	
The average time, expressed in minutes, of departure delay from all causes per flight		~	No change.	
The percentage of flights adhering to their ATFM departure slots at local level		~	Remove. The 15-minute width of the slot is too wide to be meaningful for flights with a duration of 60-150 minutes.	
Minutes of ATFM delay per delayed flight	~	~	Introduce. To be complemented by indicator on delay distribution	
Days where throughput is above expected traffic and where delays per flight are below reference value		~	Introduce. To be complemented by the number of sector hours provided.	
On time in full, OTIF	~		Introduce. Combines environment and capacity performances. Moreover, it links performance with actual service quality levels.	
Distribution of IFR flights per length of en route ATFM delay	~	~	Introduce. Provides a comprehensive view on the operational reality behind average delays per flight. Included in PRB Monitoring Reports since 2021.	
Distribution of IFR arrivals per length of air- port arrival ATFM delay	~	~	Introduce. Provides a comprehensive view on the operational reality behind average delays per flight.	
Sum of sector-opening hours	~	~	Introduce. Important to understand perfor- mance behind delays. Included in PRB Monitor- ing Reports since 2021.	

Cost-efficiency

	Scope		
Indicator	Union- wide	PRB suggestions	
Year-on-Year change of Union-wide DUC (KPI)	~		No change. To detail with indicator on cost of ca- pacity provided.
DUC at charging zone for en route and termi- nal ANS (KPI)		~	No change. To detail with indicator on cost of ca- pacity provided.
Actual unit cost incurred by users separately for en route and terminal ANS	~	~	No change.
Cost per unit of capacity	~	~	Introduce. Costs per sum of sector opening hours facilitate the understanding of interdependency between capacity and cost and would comple- ment the cost-efficiency KPI.

Network Manager – Safety

Indicator	PRB suggestions
Minimum level of the Effectiveness of Safety Manage- ment (EoSM) (KPI)	No change.
Percentage of ATFM over-deliveries	No change.
Locally identified indicators in the Network Manager man-	Introduce. EASA to define guidelines on the level of infor-
agement system	mation to be provided.

Network Manager - Environment

Indicator	PRB suggestions
The en route flight efficiency improvement generated by the European Route Network Design function related to the last filed plan trajectory (KPI)	No change.
Route extensions resulting from the actions of the Net- work Manager Operations Centre to reroute traffic to re- duce delay	Introduce. It reflects the interdependency between ca- pacity and environment by measuring the impact of de- lay-saving measures on distance flown.

Network Manager - Capacity

Indicator	PRB suggestions
Percentage of ATFM delay savings from the CDM network procedures and NM Operations Centre actions (en route and arrival ATFM delays) (KPIs)	Change. Definition in the regulation is impossible to apply. Savings to be calculated as the difference between actual ATFM delay and the original delay (i.e. the one the flight would have been subject to based on the network situa- tion).
Percentage of flights with greater than 15 minutes of en route ATFM delay	Remove, it is already included in capacity KPA.
Average daily number of ATFM regulations producing max 200 minutes of delay	Remove, no real benefit of monitoring identified.
Average en route ATFM weekend delay per flight	No change. Remove only if weighted average delay intro- duced.
Annual percentage of first rotation delay due to capacity and staffing	Change to measure all first rotation delays.
ERSA, RAI, RAU	Remove, duplicate with environment KPA.

Indicator	PRB suggestions
The unit cost for the execution of the tasks of the Net- work Manager.	No change.